Author’s response to reviews

Title: Illness recognition and care seeking for maternal complications of pregnancy and birth in rural Amhara and Oromia Regional States of Ethiopia

Authors:

Lynn Sibley (lsibley@emory.edu)
Yared Amare (yaredamare@yahoo.com)

Version: 2 Date: 05 Oct 2017

Author’s response to reviews:

Response to Editor (PRCH-D-16-00587R1)

1. Please see clarifications under Ethics approval and consent to participate (page 25)

2. Please see clarification under Consent for publication (page 25)

3. Box removed. Content presented in text (pages 20-21)

4. Please see statement in Author’s contribution section (page 20)

5. Done (page 31)

6. Noted

7. OK

8. Done